Author's response to reviews

Title: Increased tartrate-resistant acid phosphatase (Trap) expression in several malign tumors, patients sera and cell lines, an investigational study.

Authors:

Arnd Honig (arnd_hoenig@hotmail.com)
Lorenz Rieger (lrieger@gmx.de)
Michaela Kapp (M.Kapp@mail.uni-wuerzburg.de)
Mathias Krockenberger (matze.krock@web.de)
Matthias Eck (Matth.Eck@gmx.de)
Johannes Dietl (frauenklinik@mail.uni-wuerzburg.de)
Ulrike Kammerer (FRAK057@mail.UNI-wuerzburg.de)

Version: 2 Date: 15 April 2006

Dear Madam or Sir,

unfortunately I uploaded a version that contains minor language mistakes. The corrected version of the manuscript should be more convenient to read. Checking the uploaded manuscript I realized that one number in in material and methods on page 4 should be 13 and not 21.
The language and grammar changes I added are the following:

Abstract: line 1: added already and a in line 1
added In this study in line 2, line 4 In addition we investigated, line 5 Surprisingly these patients also, line 10 High values of TRAP were found in, line 11 who suffer of, line 14 markedly, line 15 test, line 17 could, line 18 might, line 19 might, line 20 + 21 not to be limited to bone affection but more general to the progress of the disease

Background: line 13 These, line 17 also, line 19 to look

Material and methods:
line 12 are, line 13 but and changed the digit into 13, page 6 chapter cell culture line 11 was reached, page 7 chapter on Immunochemistry line 10 it

Results: page 9, line 4 15 out of 21, 6 out of 8, line 5 5 out of 6, line 8 show, line 15 impressively, line 16
When investigating, page 10 line 4 who, line 5 as compared to, line 6 those of patients, line 8 progress, line 10 An analysis, line 12 showed, line 14 In comparison to, line 15/16 TRAP serum levels in later stages of gestation with greater placenta volumes tended to be higher. (better to read) line 20 the tested 5, line 22 were all TRAP, line 23 were investigated, line 26 though at, page 11 line 11 2 out of 3 line 17 we regularly

Discussion
page 12 line 11 those entities, line 12 specimens, all of the investigated, line 19 all of the investigated, line 20 has recently been, line 26 our study provides page 13 line 5 It can, line 6 also have, line 9/10 this finding is not statistically significant, line 15 still has to be, line 18 investigations have shown

Thanks a lot for the opportunity to make these stylistic and linguistic changes

yours sincerely

Arnd Honig